# Dynamic contrast-enhanced CT compared with positron emission tomography CT to characterise solitary pulmonary nodules: the SPUtNIk diagnostic accuracy study and economic modelling

Fiona J Gilbert, 1\* Scott Harris, 2 Kenneth A Miles, 1,3 Jonathan R Weir-McCall, 1 Nagmi R Qureshi, 3 Robert C Rintoul,<sup>4,5</sup> Sabina Dizdarevic,<sup>6,7</sup> Lucy Pike,8 Donald Sinclair,8 Andrew Shah,9 Rosemary Eaton,9 Andrew Clegg,10 Valerio Benedetto,<sup>10</sup> James E Hill,<sup>10</sup> Andrew Cook, 11,12 Dimitrios Tzelis, 13 Luke Vale, 13 Lucy Brindle, <sup>14</sup> Jackie Madden, <sup>11,12</sup> Kelly Cozens, <sup>11,12</sup> Louisa A Little, 11,12 Kathrin Eichhorst, 11,12 Patricia Moate, 11,12† Chris McClement, 11,12† Charles Peebles, 15 Anindo Banerjee, 15 Sai Han, 16 Fat Wui Poon, 16 Ashley M Groves, 17 Lutfi Kurban, 18 Anthony J Frew,<sup>6,7†</sup> Matthew E Callister,<sup>19</sup> Philip Crosbie,<sup>20</sup> Fergus V Gleeson,<sup>21,22</sup> Kavitasagary Karunasaagarar,<sup>23</sup> Osei Kankam<sup>24</sup> and Steve George<sup>11,12†</sup> on behalf of the SPUtNIk investigators

<sup>&</sup>lt;sup>1</sup>Department of Radiology, University of Cambridge School of Clinical Medicine, Biomedical Research Centre, University of Cambridge, Cambridge, UK

<sup>&</sup>lt;sup>2</sup>Public Health Sciences and Medical Statistics, University of Southampton, Southampton, UK

<sup>&</sup>lt;sup>3</sup>Department of Radiology, Royal Papworth Hospital, Cambridge, UK

<sup>&</sup>lt;sup>4</sup>Department of Thoracic Oncology, Royal Papworth Hospital, Cambridge, UK

<sup>&</sup>lt;sup>5</sup>Department of Oncology, University of Cambridge, Cambridge, UK

<sup>&</sup>lt;sup>6</sup>Departments of Imaging and Nuclear Medicine and Respiratory Medicine, Brighton and Sussex University Hospitals NHS Trust, Brighton, UK

<sup>&</sup>lt;sup>7</sup>Brighton and Sussex Medical School, Brighton, UK

- <sup>8</sup>King's College London and Guy's and St Thomas' PET Centre, School of Biomedical Engineering and Imaging Sciences, King's College London, London, UK
- <sup>9</sup>Radiation Protection Department, East and North Hertfordshire NHS Trust, Stevenage, UK
- <sup>10</sup>Faculty of Health and Wellbeing, University of Central Lancashire, Preston, UK
- <sup>11</sup>University Hospital Southampton NHS Foundation Trust, Southampton, UK
- <sup>12</sup>Southampton Clinical Trials Unit, University of Southampton, Southampton, UK
- <sup>13</sup>Population Health Sciences Institute, Newcastle University, Newcastle upon Tyne, UK
- <sup>14</sup>School of Health Sciences, University of Southampton, Southampton, UK
- <sup>15</sup>Department of Radiology and Respiratory Medicine, University Hospital Southampton NHS Foundation Trust, Southampton, UK
- <sup>16</sup>West of Scotland PET Centre, Gartnavel Hospital, Glasgow, UK
- <sup>17</sup>Institute of Nuclear Medicine, University College London, London, UK
- <sup>18</sup>Department of Radiology, Aberdeen Royal Hospitals NHS Trust, Aberdeen, UK
- <sup>19</sup>Department of Respiratory Medicine, Leeds Teaching Hospitals NHS Trust, Leeds, UK
- <sup>20</sup>North West Lung Centre, University Hospital of South Manchester, Manchester, UK
- <sup>21</sup>Department of Radiology, Churchill Hospital, Oxford, UK
- <sup>22</sup>University of Oxford, Oxford, UK
- <sup>23</sup>Sheffield Teaching Hospitals NHS Foundation Trust, Sheffield, UK
- <sup>24</sup>Department of Thoracic Medicine, East Sussex Healthcare NHS Trust, Saint Leonards-on-Sea, UK

\*Corresponding author fjg28@medschl.cam.ac.uk †In memoriam

**Declared competing interests of authors:** Andrew Clegg, Valerio Benedetto and James E Hill are part-funded by the National Institute for Health Research (NIHR) Applied Research Collaboration North West Coast. Fiona J Gilbert is a NIHR Senior Investigator. She was a member of the Health Technology Assessment Efficacy and Mechanisms Evaluation Board (2014–19). Andrew Cook reports grants from the NIHR Health Technology Assessment programme during the conduct of the study. Fergus V Gleeson reports other from Optellum Ltd (Oxford, UK) outside the submitted work. Robert C Rintoul reports salary support from Cambridge Biomedical Research Centre and Cancer Research UK Cambridge Centre, during the conduct of the study. Robert C Rintoul also reports personal fees from AstraZeneca and Roche for work on their Advisory Board outside the submitted work. Nagmi R Qureshi is part-funded by the Cambridge Biomedical Research Centre.

Published March 2022 DOI: 10.3310/WCEI8321

## **Plain English summary**

SPUtNIk study and economic modelling Health Technology Assessment 2022; Vol. 26: No. 17 DOI: 10.3310/WCEI8321

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

A nodule found on a lung scan cau cause concern as it may be a sign of cancer. Finding lung cancer nodules when they are small (i.e. < 3 cm) is very important. Most nodules are not cancerous. Computerised tomography (cross-sectional images created from multiple X-rays) and positron emission tomography–computerised tomography (a technique that uses a radioactive tracer combined with computerised tomography) are used to see whether or not a nodule is cancerous; although they perform well, improvements are required.

This study compared dynamic contrast-enhanced computerised tomography with positron emission tomography-computerised tomography scans to find out which test is best. Dynamic contrast-enhanced computerised tomography involves injection of a special dye into the bloodstream, followed by repeated scans of the nodule over several minutes. We assessed the costs to the NHS of undertaking the different scans, relative to their benefits, to judge which option was the best value for money.

We recruited 380 patients from 16 hospitals across England and Scotland, of whom 312 had both dynamic contrast-enhanced computerised tomography and positron emission tomography-computerised tomography scans. We found that current positron emission tomography-computerised tomography is more accurate, providing a correct diagnosis in 76% of cases, than the new dynamic contrast-enhanced computerised tomography, which provides a correct diagnosis in 70% of cases. Although dynamic contrast-enhanced computerised tomography cannot replace positron emission tomography-computerised tomography, it may represent good-value use of NHS resources, especially if it is performed before positron emission tomography-computerised tomography and they are used in combination.

Although more research is required, it may be possible in the future to perform dynamic contrast-enhanced computerised tomography at the same time as positron emission tomography-computerised tomography in patients with suspected lung cancer or if a lung nodule is found on a lung screening programme at the time of the computerised tomography examination. This may reduce the need for some people to have positron emission tomography-computerised tomography.

### **Health Technology Assessment**

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.014

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, the Cochrane Library and Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

#### **HTA** programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

#### This report

The research reported in this issue of the journal was funded by the HTA programme as project number 09/22/117. The contractual start date was in August 2012. The draft report began editorial review in December 2019 and was accepted for publication in September 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2022. This work was produced by Gilbert et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

#### NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

#### **NIHR Journals Library Editors**

**Professor John Powell** Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

**Professor Andrée Le May** Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Adviser, Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

**Professor Helen Roberts** Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk